Author's response to reviews

Title: A national survey of 'inactive' physicians in the United States of America: enticements to reentry

Authors:

Ethan A Jewett (jewett_ethan@yahoo.com)
Sarah E Brotherton (sarah.brotherton@ama-assn.org)
Holly Ruch-Ross (hruchross@aol.com)

Version: 6 Date: 11 February 2011

Author's response to reviews:

I realized that my request for the hyphen removal in "re-entry" only covered "re-entry" and not similar versions, such as "re-entered" therefore I removed the hyphen from all versions of the word. In addition, I did some formatting of the tables as there were some inconsistencies in spacing.